Author’s response to reviews

Title: Orbital sparganosis in an 8-year boy: a case report

Authors:

Xin Xie (xiexin163@163.com)
Guizhen Sun (71464166@qq.com)
Bo Ding (13858091600@163.com)
lei feng (fengleisx@126.com)

Version: 2 Date: 25 Jun 2017

Reviewer reports:

Shwu-Jiuan Sheu, M.D. (Reviewer 1): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.

It's an interesting case report, but several issues have to be solved before publication could be considered.

1. English editing is needed.
   Reply : Thanks ! We have asked a native speaker to revise the paper.

2. The information in the abstract and text should be more specific. - location of poultice applied? Conclusion - special history?
   Reply : Thanks ! We have added the requested information.

3. Line 40, add reference
   Reply : Thanks ! We have added the reference.
4. Line 45 - provide external eye photo

Reply: Thank you very much for your suggestion! We agree with the reviewer that there should be an external eye photo. However, the parents of the boy refused to permit the photo.

5. Line 95 - CT is not a specific diagnostic method in ocular sparganosis

Reply: Thank you very much for your suggestion! We revised the line 95 and also added the imaging characteristics of CT and MR imaging.